Author’s response to reviews

Title: Impact of early life exposures to geohelminth infections on the development of vaccine immunity, allergic sensitization, and allergic inflammatory diseases in children living in tropical Ecuador: the ECUAVIDA birth cohort study

Authors:

Philip J Cooper (pcooper@sgul.ac.uk)
Martha E Chico (martachico6@yahoo.es)
Irene Guadalupe (irenyta82@gmail.com)
Carlos A Sandoval (carlos_asandoval@hotmail.com)
Edward Mitre (emitre@usuhs.mil)
Tom AE Platts-Mills (tap2z@virginia.edu)
Mauricio L Barreto (mauricio@ufba.br)
Laura C Rodrigues (Laura.Rodrigues@lshtm.ac.uk)
David P Strachan (sgid950@sgul.ac.uk)
George E Griffin (ggriffin@sgul.ac.uk)

Version: 3 Date: 7 March 2011

Author’s response to reviews:

The ethical approval documents have been removed